Modelling Earth System Change: I. Validating Parameterisations for Attribution Calculations

I.G. Enting and C.M. Trudinger





Atmospheric Research

CSIRO Atmospheric Research Technical Paper no. 56

Modelling Earth System Change: I. Validating Parameterisations for Attribution Calculations

I.G. Enting and C.M. Trudinger

National Library of Australia Cataloguing-in-Publication Entry

I.G. Enting Modelling Earth System Change: I. Validating Parameterisations for Attribution Calculations

Bibliography. ISBN 0 643 06652 7

CSIRO Atmospheric Research Technical paper ISSN 1445-6982

 Greenhouse gases – Mathematical models. 2 Greenhouse effect, Atmospheric – Mathematical models.
 Climatic changes – Mathematical models. I. Trudinger, Catherine Mary. National Library. II. CSIRO. Division of Atmospheric Research. III. Title. IV Title: Validating parameterisations of attribution calculations. (Series: CSIRO Atmospheric Research technical paper; no. 56).

551.5253011

Address and contact details: CSIRO Atmospheric Research Private Bag No.1 Aspendale Victoria 3195 Australia Ph: (+61 3) 9239 4400; fax: (+61 3) 9239 4444 e-mail: ar-enquiries@csiro.au

CSIRO Atmospheric Research Technical Papers may be issued out of sequence. From July 2000, all new Technical Papers will appear on the web site of CSIRO Atmospheric Research. Some Technical Papers will also appear in paper form.

Modelling Earth System Change: I. Validating Parameterisations for Attribution Calculations

I.G. Enting and C.M. Trudinger CSIRO Atmospheric Research Private Bag 1, Aspendale, Vic 3195, Australia

Abstract

This paper describes a set of analyses to validate simple model parameterisations as part of a model assessment for calculations attributing global change to particular groups of emissions. The report also identifies issues involved for the assessment to address the recommendations from the SBSTA experts' meeting (Bonn, May 2001). DISCLAIMER: The scenarios for future emissions used in this report were chosen to provide a standard case for the various models in the intercomparison. This case should not be treated as having any special status as a likely future. In particular, it should be noted that the SRES A2-marker case is predicated on a global population of around 15 billion in 2100.

1 Introduction

This report documents a set of standardised calculations that are intended as validation of simple parameterised models of global change.

The SBSTA assessment, for which the present study was undertaken, arose from a proposal from Brazil to set emission targets on the basis of historical responsibility for global change. The proposal was presented to the Ad Hoc Group on the Berlin Mandate (AGBM) and subsequently referred to the Subsidiary Body for Scientific and Technical Advice (SBSTA) that was set up under the U.N. Framework Convention on Climate Change (FCCC). A summary of the formal consideration within the FCCC process (through to 2001) is given by Enting and Law (2002) (see also SBSTA, 2001).

Our modelling of global change using parameterised models is part of an on-going research activity in CSIRO (Australia). The methodology involves developing highly-modularised models so that we can, as desired, investigate particular aspects of global change, using detailed modelling of individual system components with inputs and/or outputs from/to less detailed representations of other components of the earth system. The present report is intended as the first in a series of CSIRO Atmospheric Research Technical Papers with the common title: *Modelling Earth System Change* and numbered I, II, etc. in addition to the CAR Technical paper numbers.

The SBSTA intercomparison involved the following components:

- A 'reference' case, supplied by SBSTA, based mainly on analyses with the U.K. Hadley Centre HadCM3 Atmosphere-Ocean General Circulation Model (AOGCM).
- A default case, comprising a set of standard inputs (emissions of CO₂, CH₄ and N₂O; direct and indirect radiative forcing for aerosols) and a set of standard parameterisations for: the dependence of concentrations on emissions, the dependence for forcing on concentration and the dependence of temperature (and sea-level) on forcing.
- A comparison between this default case and the reference case, to test that simple parameterisations could reproduce the behaviour of complex models. To the extent that the default is fully specified, all modelling groups that use the default should obtain identical results (to numerical accuracy). There are however, a few details that were not specified in the default case. These are noted in the description below.
- Similar comparisons with alternative parameterisations using either the same inputs as the default case or alternative estimates. We will define a 'preferred' case, representing our first step in refining the default case.

The layout of the remainder of this report is as follows. Section 2 describes the 'default' case. Section 3 describes various extensions that we consider for CO_2 . Section 4 describes the other gas concentrations. Section 5 describes the validation of the climate change variables. Section 6 revisits these various results, relating changes to time of emission. Section 7 reviews the recommendations of the Bonn SBSTA experts' meeting and Section 8 issues involved in sensitivity analyses.

2 Default parameterisations

For the purposes of intercomparison of models, a number of standard representations have been defined. These are:

- standard forcings, expressed as emissions, concentrations or radiative forcing, depending on the constituent;
- standard default responses of particular components of the earth system.

The present section describes these standard cases. Modellers who use all the default responses should (to within numerical accuracy) all obtain the same results, apart from small differences due to some aspects of the default being left unspecified. Of course, the aim of the assessment is to determine the extent to which the results differ when different parameterisations are used.

Inputs:

- **Past CO**₂ **emissions** From CDIAC, as the sum of fossil carbon emissions (Marland et al., 2001) and emissions from land-use change (Houghton, 1999; Houghton and Hackler, 2000).
- Past CH₄ emissions From EDGAR-HYDE data set (van Aardenne et al., 2001).
- Past N₂O emissions From EDGAR-HYDE data set (van Aardenne et al., 2001).
- Future CO₂ emissions SRES A2-marker scenario. (Nakićenović et al, 2000: p471).
- Future CH₄ emissions SRES A2-marker scenario (Nakićenović et al, 2000: p471).
- Future N₂O emissions SRES A2-marker scenario (Nakićenović et al, 2000: p471).
- **Direct radiative forcing of aerosols** Prescribed function from SRES A2-marker scenario, see Figure 2.
- **Indirect radiative forcing of aerosols** Prescribed function from SRES A2-marker scenario, see Figure 2.

The emission estimates and scenarios are presented as annual values for specific years. For computational convenience, these are actually used as instantaneous values in the middle of the year, with linear interpolation used for other times. The SRES A2 scenario for gas emissions is given in table 1.

Model relations:

- CO_2 concentrations Perturbation from pre-industrial concentration of 280 ppm. Response given by Bern model expression, equation (2.1).
- **CH**₄ **concentrations** Perturbation from pre-industrial concentration of 700 ppb; fixed 8.4 year lifetime; (Prather et al., 2001: Table 4.1).
- N_2O concentrations Perturbation from pre-industrial concentration of 275 ppm; fixed 120 year lifetime; (Prather et al., 2001: Table 4.1).
- **Radiative forcing of gases** Functional relations from the IPCC Third Assessment Report, (equations 2.4a–h).
- Warming Response function from HadCM3, equation (2.5a).

Sea level rise Response function from HadCM3, equation (2.6a).

The radiative forcing, warming and sea-level rise are all expressed as differences from preindustrial conditions. The emissions are the anthropogenic inputs that have caused these changes. Therefore the concentrations calculated using the response functions are perturbations from pre-industrial. Comparisons with observed concentrations will require specification of the preindustrial value. The default calculations use the values specified above, which are taken from the IPCC Third Assessment Report (IPCC, 2001).

Component	Emissions	Concentrations	Forcing
CO_2	CDIAC	280 ppm + Bern response	Eq. 2.4c
CH_4	EDGAR-HYDE	700 ppb + 8.4 yr lifetime	2.4d
N_2O	EDGAR-HYDE	275 ppb + 120 yr lifetime	2.4e
aerosol (direct)	—		prescribed
aerosol (indirect)	—		prescribed
halocarbons	—		

These default specifications can be summarised as:

The default response for CO_2 is the Bern model response from the Third Assessment Report, expressed as

$$R_{\text{Bern:TAR}}(t) = 0.152 + 0.253 \exp(-t/171.0) + 0.279 \exp(-t/18.0) + 0.316 \exp(-t/2.57)$$
(2.1)

The gas response functions are defined to have R(0) = 1, and so must be used with a scale factor, giving

$$C_{\rm CO2}(t) = C_{\rm CO2:eqm} + A_{\rm CO2} \int_{t_0}^t R(t - t') E_{\rm CO2}(t') dt'$$
(2.2)

where the factors A_{η} define the conversion from emission units (generally in terms of masses) to gas units (usually mixing ratios).

For gases that can be characterised by a single lifetime, τ_{η} , we have

$$R_{\eta} = \exp(-t/\tau_{\eta}) \tag{2.3}$$

Our calculations work in the units used for the emissions: Pg carbon (Gt C) for CO₂; Tg CH₄ (Mt CH₄) for CH₄; Tg N (Mt N) for N₂O. These are the units generally used in IPCC reports. Note that the EDGAR-HYDE CH₄ emissions are in Tg C y⁻¹, not Tg CH₄ y⁻¹ (van Aardenne et al., 2001).

We use the conversion factors:

$A_{\rm CO2}$	= 0.471	ppm/(Pg C)	ppm/(Gt C)
$A_{\rm CH4}$	$=\frac{12}{16}A_{\rm CO2}$	ppb/(Tg CH ₄)	ppb/(Mt CH ₄)
$A_{\rm N2O}$	$= \frac{12}{28} A_{\rm CO2}$	ppb/(Tg N)	ppb/(Mt N)

These factors come from the combination of the total (dry) mass of the atmosphere, the mean molecular weight of the atmosphere and the molecular weight of the species.

The specifications prescribe the use of radiative forcing expressions from the IPCC Third Assessment Report. The radiative forcing is expressed as a sum over contributions:

$$F(t) = F_{\rm CO2}(t) + F_{\rm CH4}(t) + F_{\rm N2O}(t) + F_{\rm aerosol}(t) + \dots$$
(2.4*a*)

(An overlap contribution for CH_4 and N_2O is included in the forcing for these gases).

The individual forcings are expressed as functions of the concentration:

$$F_{\eta}(t) = f_{\eta}(C_{\eta}(t)) \tag{2.4b}$$

For CO₂ we use

$$f_{\rm CO2} = 5.35 \ln(C_{\rm CO2}/C_{\rm CO2:eqm}) \tag{2.4c}$$

where $C_{\text{CO2:eqm}}$ is the notional pre-industrial equilibrium concentration.

This form is quoted by the IPCC (Ramaswamy et al., 2001: Table 6.2) as one of several alternatives. Other functions are shown in Figure 1, including the older expression previously used by the IPCC (shown by the uppermost line in Figure 1) in which the factor 6.3 was used instead of 5.35.

$$f_{\rm CH4}(C_{\rm CH4}) = 0.036 \left[\sqrt{C_{\rm CH4}} - \sqrt{C_{\rm CH4:eqm}} \right] - f_{\rm overlap}(C_{\rm CH4}, C_{\rm N2O:eqm}) + f_{\rm overlap}(C_{\rm CH4:eqm}, C_{\rm N2O:eqm}) \quad (2.4d)$$

$$f_{\rm N2O}(C_{\rm N2O}) = 0.12 \left[\sqrt{C_{\rm N2O}} - \sqrt{C_{\rm N2O:eqm}} \right] - f_{\rm overlap}(C_{\rm CH4:eqm}, C_{\rm N2O}) + f_{\rm overlap}(C_{\rm CH4:eqm}, C_{\rm N2O:eqm}) \quad (2.4e)$$

Equation (2.4e) represents a change from the function previously used by the IPCC, which had a coefficient of 0.14 rather than 0.12. The CH_4 -N₂O overlap contribution is given by:

$$f_{\text{overlap}}(C_{\text{CH4}}, C_{\text{N2O}}) = 0.47 \ln[1 + 2.01 \times 10^{-5} (C_{\text{CH4}} C_{\text{N2O}})^{0.75} + 5.31 \times 10^{-15} C_{\text{CH4}} (C_{\text{CH4}} C_{\text{N2O}})^{1.52}] \qquad (2.4f)$$

where in (2.4d,e,f) C_{CH4} and C_{N2O} are in ppb. We also follow the IPCC (Shine et al., 1990: Table 2.2) and include an indirect effect of methane as a contribution from stratospheric water vapour given by

$$f_{\rm CH4:H2O} = 0.011 [\sqrt{C_{\rm CH4}} - \sqrt{C_{\rm CH4:eqm}}]$$
 (2.4g)

For other gases, the forcing relation can be treated as linear and expressed in the form:

$$f_{\eta} = a_{\eta} C_{\eta} \tag{2.4h}$$

where the IPCC Third Assessment Report quotes $a_{CFC-11} = 0.25 \text{ W m}^{-2} \text{ (ppb)}^{-1}$ and $a_{CFC-12} = 0.32 \text{ W m}^{-2} \text{ (ppb)}^{-1}$.

The HadCM3 warming was specified as the response to a delta-function forcing as:

$$W_{\text{default}}(t) = 0.0744423 \exp(-t/8.4007) + 0.001038 \exp(-t/409.54)$$
(2.5a)

which is used in the form expressing the temperature increase, T(t), in terms of the forcing, F(t), as

$$T(t) = \int_{t_0}^t W(t - t')F(t') dt$$
(2.5b)



Figure 1: Four alternative expressions for the radiative forcing of CO_2 as a function of concentration as noted by Ramaswamy et al. (2001: Table 6.2). Short dashes are the form from Shine et al. (1990) used by Enting and Law (2001). The longer dashes give an revised representation. Solid curve is a simpler approximation, given by equation (2.4c), and used in present work. Chain curve is a function used in earlier WMO studies.

Response function representations of a number of other climate models are listed by den Elzen et al. (1999: Table 3.4).

Similarly, the HadCM3 sea-level rise was specified as the response to a delta-function forcing expressed as:

$$V_{\text{default}}(t) = 6.659 \times 10^{-4} \exp(-t/1700.2) + 3.85 \times 10^{-7} \exp(-t/33.788)$$
(2.6a)

which is used in the form giving the sea-level rise, S(t), as

$$S(t) = \int_{t_0}^t V(t - t') F(t') dt$$
(2.6b)

The results of calculations using these default specifications are shown in the various figures as the 'Default' curves. Concentrations for the historical period are shown in Figures 3, 5 and 7 for CO_2 , CH_4 and N_2O respectively. The corresponding results for 1800–2100 (i.e. with the default SRES A2-marker scenario) are shown in Figures 4, 6 and 8. Figure 9 shows only results from the default case and gives the radiative forcing and its contributions from CO_2 , CH_4 , N_2O and the two aerosol contributions. Figure 11 gives the warming for 1800–2100.



Figure 2: Prescribed aerosol forcing. Future values based on A2 'SRES' scenario. The smaller magnitude component is direct forcing, the larger magnitude is indirect forcing.

t	CO_2	CH_4	N_2O
yr	$Pg C y^{-1}$	$Tg CH_4 y^{-1}$	$Tg N y^{-1}$
1990.5	7.10	310.0	6.7
2000.5	7.97	323.0	7.0
2010.5	9.58	370.0	8.1
2020.5	12.25	424.0	9.6
2030.5	14.72	486.0	10.7
2040.5	16.07	542.0	11.3
2050.5	17.43	598.0	12.0
2060.5	19.16	654.0	12.9
2070.5	20.89	711.0	13.9
2080.5	23.22	770.0	14.8
2090.5	26.15	829.0	15.7
2100.5	29.09	889.0	16.5

Table 1: SRES A2-marker scenario (Nakićenović et al., 2000: p471).

To summarise, the choices for the 'default' for aspects where the specifications were incomplete are:

- In the absence of specified initial conditions we use 280 ppm for CO_2 , 700 ppb for CH_4 and 275 ppb for N_2O , based on the IPCC Third Assessment Report.
- In the absence of specified default for historical emissions of CH_4 and N_2O , we have used the EDGAR-HYDE estimates (van Aardenne et al., 2001). These have been back-extrapolated (linearly) to zero in 1750.
- We have also used a linear back-extrapolation to zero in 1750 for CO₂ emissions from land-use change.
- We have included an indirect contribution to radiative forcing from stratospheric H_2O produced *in situ* by oxidation of CH_4 (Shine et al., 1990).

3 Validation of CO₂ concentrations

3.1 Background

An important emphasis in this report is on exploring alternative parameterisations of the carbon cycle. This is motivated by:

- results such as Enting et al. (1994) concluding that budget uncertainty is the main contribution to uncertainties in projections of future CO₂ concentrations (for a given emissions scenario);
- den Elzen and Schaeffer (2000: Fig. 4d) finding that carbon budget uncertainty (specifically the role of deforestation) was the main contribution to uncertainty in attribution of relative responsibility for the warming to 1990.

Our modelling framework is being developed to build on our work on uncertainty analysis:

- work on treating the emissions as time series whose values and uncertainties are to be estimated (Trudinger, 2000; Trudinger et al., 2002a,b);
- our approach to capturing model uncertainty through a Bayesian calibration (Enting and Pearman, 1987; Enting and Lassey, 1993; Trudinger, 2000);
- the work by Enting and Trudinger (2001) characterising uncertainties in CO₂ projections in terms of the Laplace Transforms of model responses.

Over the historical period, the primary comparison is with the concentrations observed either directly or recovered from gas trapped in polar ice. However, as can be seen in Figure 3, for CO_2 the reference case follows this curve closely. For the 21st century, the comparison must, of course, simply compare the simple models to the reference case.

The comparisons of the concentrations of CO_2 over the historical period (Figure 3) and for the period to 2100 using the SRES A2-marker scenario (Figure 4) indicate a number of discrepancies. On the whole, these are to be expected on the basis of current understanding:

- i for CO_2 , the anomaly around 1940 is not explicable on the basis of the emissions estimates. This has been further studied by Trudinger (2000) who noted that the ice-core record represents a smoothing of the atmospheric record. The flux variations that caused the anomaly must have been of larger amplitude than would be implied if the ice-core record was treated as an unsmoothed record of atmospheric concentration.
- ii the known non-linearities in the response of CO₂ (see Section 3.3 below) mean that response functions linearised about pre-industrial conditions will give a poor representation at high CO₂ concentrations (e.g. Joos et al., 1996). Therefore the increasing discrepancy over the 21st century is to be expected.

3.2 Budget and calibration

As noted above, current carbon budget uncertainty has been identified as a major contributor to uncertainties in both projections of future CO_2 and attribution of responsibility. Carbon budget uncertainty has generally been addressed in terms of the air/sea partitioning. However, den Elzen and Schaeffer (2001) noted that it was uncertainties in deforestation (i.e. uncertainties in the partitioning between forcing and response) that made the greatest contribution to uncertainty in attribution. In this context, it is important to consider what Enting (2002: Fig. 14.2) termed the 'functional role' of the carbon fluxes.

This issue is illustrated by comparing two alternative views of the so-called 'missing carbon sink'.

- i The missing sink is an aspect of natural variability (as, for example, explored by Dai and Fung, 1993). Regardless of any systematic trend over recent years, the best estimate for the future behaviour of these processes is taken as zero. This is the assumption underlying a number of the calculations reported in the 1990 IPCC report (e.g. Enting, 1991).
- ii The missing sink is due to processes linked to global change and this linkage will continue into the future. On this basis, the process is often modelled as CO₂-fertilisation, recognising that this may be a proxy for a range of processes such as nitrogen fertilisation. This modelling approach was adopted for the set of model runs reported by Enting et al. (1994) which contributed to the IPCC Radiative Forcing Report and the Second Assessment Report.

For the present study, the 'functional role' classification from Enting (2002) can be re-expressed as:

- **direct anthropogenic forcing** These are the emissions that result from human action without a causal dependence on the atmospheric concentration.
- **response to anthropogenic forcing** These are the fluxes that change in response to anthropogenic changes in atmospheric concentration.
- natural variation This will be things such as flux variations driven by climate variability as well as 'internal' variation due to such things as disturbance and succession in terrestrial ecosystems and circulation/production variations in the oceans. Enting (2002: Section 6.1) noted that the extent to which these variations should be modelled as deterministic vs. stochastic processes will depend on what other information is available.

Wigley and Raper (1992) presented calculations comparing the two budgeting approaches described above. They termed them the 'no-feedback' and 'feedback' cases respectively and claimed that case (ii) represented balanced carbon budgets while (i) represented unbalanced budgets. Both aspects of this terminology are misleading and inappropriate. As indicated above, the difference lies in how the budget is balanced, not whether it is balanced. Since CO_2 -fertilisation represents a direct response to increased atmospheric CO_2 (as does air-sea gas exchange) there seems to be little scientific or semantic justification for describing CO_2 fertilisation as a 'feedback', and real potential for confusion with the influence of climate change on the carbon budget.

Wigley and Raper (1992: Fig 1) indicate that the difference in the approach to budgeting will lead to changes of order 50 to 80 ppm in CO_2 concentrations projected for 2100 with a given scenario. Note however, that the specific values given by Wigley and Raper incorporate an error that effectively leads to double-counting of the recovery from land-use change (Enting et al., 1994: Technical note A.6.B).

Figure 3 shows a general agreement between the observations and model, apart from the period around 1940. For more detailed comparisons we can follow the Enting et al. (1994) study and compare the 1980s' budget in more detail.

As well as the 'default' response, and our 'preferred' representation (described in Section 3.3) we consider two additional responses:

$$R_{\text{Bern:SAR}}(t) = 0.1369 + 0.1298 \exp(-t/371.6) + 0.1938 \exp(-t/55.70) + 0.2502 \exp(-t/17.01) + 0.2086 \exp(-t/4.16) + 0.0807 \exp(-t/1.33)$$
(3.2.1)

specified as an option in the SBSTA intercomparison, and

 $R_{\text{Bern:INIT}}(t) = 0.130164 + 0.333279 \exp(-t/4.144656) + 0.260540 \exp(-t/18.587414)$

 $+0.165742 \exp(-t/58.455562) + 0.110275 \exp(-t/414.962281)$ (3.2.2) quoted by Enting et al., (1994: Table 9.4).



Figure 3: Historical CO_2 concentrations, comparing: reference case (dotted line); calculated with default emissions and responses (solid line); calculated with preferred emissions and responses (dashed line); Law Dome ice core data (Etheridge et al., 1996) (open circles).

The comparisons for 1980 to 1990 are:

Case	Response	C_0	C(1980.5)	C(1990.5)	growth
	(Eqn.)	ppm	ppm	ppm	$ppm y^{-1}$
Obs.		282.5			1.53
Default	Bern:TAR (2.1)	280	343.5	359.4	1.59
	Bern:SAR (3.2.1)	280	340.6	356.5	1.59
	Bern:INIT (3.2.2)	280	338.3	355.4	1.51
Preferred	Various	280	336.8	352.2	1.54

Ideally, one would like to be able to use this type of comparison as a way of choosing between simple models. There are, however, a number of aspects that limit what can be achieved:

- The problem of using emissions and concentrations to estimate the model response is ill-conditioned. Given a relation C(t) C₀ = ∫ R(t t')E(t') dt', deconvolution for R(.) given C(.) and E(.) is mathematically the same as the 'standard' deconvolution for E(.) given C(.) and R(.) (Enting, 2002: Section 6.6).
- The difficulty in assessing uncertainty can be characterised as one of finding the appropriate set of basis functions to characterise the uncertainty. This has some analogies to the

aggregation problem encountered in synthesis inversion (Kaminski et al., 2001; Enting, 2002: Section 8.3).

• A further difficulty is that emissions are not exactly known. In the worst case, this can lead to circularity (and under-estimation of errors) if models are 'validated' with model-derived emission estimates, and at best one is faced with a complex estimation problem.

Some potential solutions to these difficulties can be identified:

- Enting (2002: Box 8.1) quotes Anderssen (1977) as noting that sometimes one does not actually want the solution of such inverse problems, what is required is some linear functional of the solution. In such cases, Anderssen recommends using techniques that calculate the functional directly.
- As a specific example, Enting and Trudinger (2001) characterised uncertainties in projections of future CO₂ in terms of airborne fractions. Using the expression for airborne fraction in terms of the Laplace Transform of the response (e.g. Enting, 1990), they suggested that uncertainties in future projections could be parameterised in terms of how accurately the airborne fraction could be estimated over a calibration period and the sensitivity of the airborne fraction to changes in emission growth rate.
- In analogy to the Enting and Trudinger analysis, we can expect that relative responsibilities for the greenhouse effect will depend on two characteristics: (a) the relative levels of emissions over the attribution period (b) differences in the time distribution within the period. The contribution to *relative* responsibility from (a) should be insensitive to intermodel differences but differences in relative responsibility associated with different time distributions of emissions can be expected to be sensitive to inter-model differences. The work of den Elzen and Schaeffer (2000, 2002) suggests that this contibutes less than 2% uncertainty in the attribution to Annex-1 vs. others, but that at the level of individual nations larger differences occur for nations with rapidly-growing emissions.

A number of proposals have been made for *ad hoc* tuning of response functions:

- The specifications for the SBSTA intercomparison suggest that this should be done via the A_n factors.
- An alternative (Wigley, 1991) is to modify response functions by (linear) transformation of the time argument.

However, in adopting such tuning, the important point is to ensure that the formalism does not inappropriately reduce the uncertainty. In terms of the discussion above, the tuning should not restrict the range of variation of the linear functionals that reflect attribution, unless such a restriction is inherent in the data used for tuning.

3.3 Non-linearity

An important complication in analysing attribution of the greenhouse effect is the non-linearity in the response of CO_2 from emissions. This non-linearity arises from two main causes:

- non-linearity in the relation between carbon content and CO₂ partial pressure in the ocean surface layer;
- non-linearity (and likely saturation) of the CO₂-fertilisation effect in the terrestrial biota.

The limited validity of linear response function descriptions of CO_2 has long been known. Examples explicitly noting different behaviour at higher concentrations are given by Maier-Reimer and Hasselmann (1987) for ocean response functions and Enting et al. (1994) who reported three different response functions for the Bern model under conditions of perturbations from (a) pre-industrial (equation 3.2.2); (b) fixed 1990 emissions; (c) the S650 stabilisation scenario.

Because these non-linear effects are most important for future concentrations, there are only inter-model comparisons possible. Historical data are little-influenced by the non-linearity. The difficulty is particularly acute for non-linearity in CO_2 fertilisation. In comparison, for air-sea gas exchange the non-linearity is weaker and the behaviour can be predicted from a combination of direct chemical measurements and calculations of chemical equilibrium.

A number of computational techniques have been developed to represent the non-linearity:

- Wigley (1991) interpolated between response functions reported by Maier-Reimer and Hasselmann (1987);
- Joos et al. (1996) proposed explicit modelling of air-sea CO₂ exchange and a response function describing the ocean mixed-layer. This approach also requires a model of the terrestrial carbon response.

To further quantify the significance of these non-linearities, an additional calculation was performed using the IS92a emission scenario. Using response $R_{\text{Bern:TAR}}$ gave a concentration of 625.8 ppm in 2100. This is lower than results reported by Enting et al. (1994). The models in that study gave ranges, clustered in two groups, around 685 ppm and 725 ppm. This confirms that, as expected from response functions describing perturbations about pre-industrial conditions, $R_{\text{Bern:TAR}}$ gives results that are too small when applied to larger emissions over the 21st century.



Figure 4: Projected CO_2 concentrations, for SRES A2-marker scenario. reference case (dotted line); calculated with default emissions and response (Solid line); calculated with our preferred representation based on explicit non-linear expressions for air-sea exchange and NPP (dashed line).

3.4 The 'preferred' CO₂ case

Our 'preferred' representation of CO_2 is essentially that described by Joos et al (1996). It has an explict (non-linear) expression for the flux, Φ_{NPP} from Net Primary Production (NPP) of the terrestrial biota, an explicit (weakly non-linear) expression for air-sea gas exchange flux Φ_{AS} , a response function describing mixed-layer concentrations responding to air-sea exchange, a response function describing how the respiration and decay in the terrestrial biota responds to changes in NPP. The model expresses the rate of change of atmospheric CO_2 content as emissions minus NPP plus respiration minus air-sea exchange. The specific equations are:

For the NPP:

$$\Phi'_{\rm NPP} = \Phi_{\rm NPP}(C) - \Phi_{\rm NPP:eqm} = \alpha \Phi_{\rm NPP:eqm} \frac{(G_{\infty} - 1)(C - C_0)}{G_{\infty}(C_0 - C_c) + (C - C_0)}$$
(3.4.1)

following Enting and Lassey (1993) in the use of the expression from Allen et al. (1987). Here C_c is the concentration at which photosynthesis ceases, taken as 80 ppm, G_{∞} is the limiting enhancement, taken as 2.4 and $\Phi_{\text{NPP:eqm}}$ the equilibrium NPP, is taken as 100 GtCy⁻¹. Notionally, the factor α respresents the fraction of the terrestrial NPP to which this enhancement applies. It was used as a tuning parameter by Enting and Lassey.

The air-sea gas exchange is expressed in terms of an atmospheric turnover time κ^{-1} as:

$$\Phi_{\rm AS} = \kappa \left[M_{\rm a} - M_{\rm a:eqm} - \xi M'_{\rm m} M_{\rm a:eqm} / M_{\rm m:eqm} \right]$$
(3.4.2*a*)

where $M_{\rm a}$ and $M_{\rm m}$ are the atmospheric and mixed-layer carbon contents and $M_{\rm a:eqm}$ and $M_{\rm m:eqm}$ are their respective equilibrium values.

Putting

$$y = M'_{\rm m}/M_{\rm m:eqm} \tag{3.4.2b}$$

the 'buffer factor', ξ , is (Bacastow, 1981)

$$\xi(y) = 9.36 + 59.56y + 4558y^3 \tag{3.4.2c}$$

The response of the mixed-layer to air-sea flux is expressed as:

$$M'_{\rm m}(t) = \int R_{\rm mixed}(t - t')\Phi_{\rm AS}(t')\,dt'$$
(3.4.3)

and the atmospheric budget is written as

$$\frac{d}{dt}M_{\rm a} = E(t) - \Phi_{\rm AS} - \Phi_{\rm NPP} + \int R_{\rm decay}(t-t')\Phi_{\rm NPP}(t')\,dt' \tag{3.4.4}$$

The Bern model biosphere model has

$$R_{\text{decay:Bern}} = 0.70211 \exp(-0.35t) + 0.013414 \exp(-t/20)$$
$$-0.71848 \exp(-55t/120) + 0.0029323 \exp(-t/100) \qquad (3.4.5a)$$

However, for the present, we use a simple 2-box form:

$$R_{2-\text{box}} = 0.8\lambda_1 \exp(-\lambda_1 t) + 0.2\lambda_2 \exp(-\lambda_2 t)$$
(3.4.5b)

with $\lambda_1^{-1} = 2$ yr and $\lambda_2^{-1} = 50$ yr. This approximates the behaviour of the two-reservior biosphere used in our work with the box-diffusion model (e.g. Enting and Lassey, 1993; Trudinger, 2000). With this response function, balanced carbon budgets (balanced according to option (ii) of Section 3.2) correspond to NPP enhancements with $G_{\infty} \approx 2.4$ and $\alpha \approx 1$. Our numerical studies have found that the balance between Φ'_{NPP} and the respiration/decay response are very sensitive to the form of the latter. This needs to be carefully considered in the uncertainty analyses for attribution calculations. Studies of terrestrial response, such as those of Meyer et al. (1999) and Thompson and Randerson (1999), may provide a useful starting point for refining our model.

In describing mixed-layer responses, Joos et al. (1996) used distinct representations for short times and times longer than about a year. For the present, we use a single representation based on the long-time form of the HILDA model response from Joos et al. with the amplitude of the fastest decay rescaled to give $R_{\text{mixed}}(0) = 1$. This should be an acceptable approximation for the present studies but it may be unsuitable for isotopic studies.

$$R_{\text{mixed}} \approx 0.022936 + 0.674747 \exp(-t/1.2679) + 0.13963 \exp(-t/5.2528)$$

 $+0.089318 \exp(-t/18.601) + 0.037820 \exp(-t/68.736) + 0.035549 \exp(-t/232.3)$ (3.4.6)

This formalism defines our 'preferred' case for CO₂, with the parameters tuned to $\alpha \Phi_{\text{NPP:eqm}} = 88 \text{ Gt C y}^{-1}$. Figure 3 shows that this representation gives a good fit to the observational data and Figure 4 shows it reproduces the behaviour of the reference case more accurately than would be expected from the spread of models reported by Enting et al. (1994). For comparison with this earlier work, we found that the IS92a scenario gave a CO₂ concentration of 679.5 ppm in 2100, confirming that our preferred case gives much better agreement with other carbon cycle models than does the default linear response. We found that, with this non-linear response, we could improve the agreement with 1980s' concentrations by following Trudinger (2000), and starting our integrations from the value 282.5 ppm. This corresponds to concentrations from before the Little Ice Age. In other words, the Little Ice Age would be treated as an anomaly with little impact on the post-1800 CO₂ record. This is consistent with ¹³C data from ice cores.

4 Validation of concentrations of other greenhouse gases

4.1 Methane

The comparisons of the concentrations of methane over the historical period (Figure 5) and for the period to 2100 using the SRES A2-marker scenario (Figure 6) indicate a number of discrepancies. As with CO_2 , these are to be expected on the basis of current understanding:

- i the EDGAR methane budget is consistent with older estimates of order 10 years for the methane lifetime and this is inevitably inconsistent with the newer estimate of 8.4 years.
- **ii** Over the 21st century, the 8.4 year lifetime leads to increasing discrepancies from the reference case, a longer lifetime being needed. This is expected because of methane's effect on its own lifetime, although quantitative modelling would have to include the effects of other gases (e.g. CO) on the methane lifetime. Analysing the budget of the reference case indicates a lifetime of 10.1 years at the end of the 21st century.

For our 'preferred' representation of CH_4 we use the initial refinement of a time-varying lifetime.

$$\lambda^{-1} = 8.4 \qquad \text{for } t \le 1900 \tag{4.1.1a}$$

$$\lambda^{-1} = 8.4 + 0.8(t - 1900)/100 \quad \text{for } t > 1900 \quad (4.1.1b)$$

In our modelling, this is applied only to the perturbation and not to the natural sources. Therefore it corresponds to an integral form of the perturbation lifetime (or effective lifetime) defined by Prather (1994) (see also Enting, 2002: equations 15.1.1a–g). Figure 6 shows that the use of



Figure 5: Historical CH_4 concentrations. reference case (dotted curve); calculated with default emissions and response (solid curve); preferred case, using linearly-varying lifetime after 1900 (dashed curve); Law Dome ice core data (Etheridge et al., 1998) (open circles).

the time-varying lifetime gives results over the 21st century that are significantly closer to the reference case than those given by the default.

The modelled concentrations are higher than the ice-core data. An offset is to be expected because of the spatial gradients in methane concentrations. However, the offset shown in Figure 5 is larger than that implied by spatial gradients (Etheridge et al., 1998).

In defining a 'preferred case' our aim is to have a parameterisation that can be used in a consistent way with the other model components, so that changes in components can be explored independently. However, in exploring the significance of any model component, we would usually expect to go beyond the parameterisations of our preferred case *for that component*. In such studies we would specify the behaviour of the other components using their respective preferred cases.

4.2 Nitrous oxide

The comparisons of the concentrations of nitrous oxide over the historical period (Figure 7) and for the period to 2100 using the SRES A2-marker scenario (Figure 8) indicate a number of discrepancies. As with CO_2 and CH_4 , these are to be expected on the basis of current understanding:



Figure 6: Projected CH_4 concentrations, for SRES A2-marker scenario. reference case (dotted curve); calculated with default emissions and response (solid curve); preferred case, using linearly-varying lifetime after 1900 (dashed curve).

- i The EDGAR-HYDE emission estimates are too small to match the observed concentrations. The back-extrapolation to 1750 improves the agreement slightly (since the long lifetime means that the effect of early emissions persists). However the main cause of the discrepancy is that the EDGAR-HYDE estimates for 1990 are 2.85 Tg N y⁻¹ while the IPCC Third Assessment Report (Prather et al., 2001) indicates that atmospheric budget analysis implies an anthropogenic N₂O source of 5.7 Tg N y⁻¹.
- ii Comparison with the reference case suggests that the lifetime decreases over the 21st century. This is understood in terms of other changes in the chemical composition of the atmosphere (e.g. Prather et al. 2001, and references therein). However analysing the budget of the reference case indicates a lifetime of about 140 years at the end of the 21st century. Compared to the reference case, the default and preferred cases seem to have lifetimes that are too long early in the 21st century.

For our 'preferred' representation of N_2O we use a deconvolution of ice-core data. The emissions over the historical period are calculated as

$$E_{\rm N2O} = \left[\frac{d}{dt}C_{\rm N2O} + C_{\rm N2O}/120.0\right]/A_{\rm N2O}$$
(4.2.1)

Since these emissions are derived directly from observed concentrations, comparisons with concentration data do not provide any validation of the model. In this case, the circularity



Figure 7: Historical N_2O concentrations, comparing: reference case (dotted curve); calculated with default emissions and response (solid curve); preferred case, based on spline fit to ice core data (dashed curve); selected ice-core and firn data (open circles) from Machida et al. (1995) and Battle et al. (1996).

involved in using concentration data to validate emission estimates is clear. In cases where various forms of tuning have been applied, the circularity can still be present but less obvious.

The incomplete understanding poses a problem for attribution calculations. The 'preferred' case here represents a 'worst case' making use of no independent source estimates of N_2O over the historical period. While this can, no doubt, be improved, uncertainties in the global budget inevitably imply uncertainties in attribution of components.

4.3 Halocarbons

The halocarbons bring additional complications. The Kyoto Protocol explicitly excludes gases whose emissions are restricted by the Montreal Protocol (including its various amendments). However, since the Brazilian proposal aims to go beyond the Kyoto Protocol, there is no a priori reason for excluding the halocarbons. Currently, the combined radiative forcing from the halocarbons lies between that from methane and that from nitrous oxide. Clearly there is an element of responsibility for the emissions, and compared to other minor gases, the emissions data are of relatively high quality, having been assembled because of the ozone-depleting role of these compounds. Therefore, inclusion of the halocarbons is likely to increase the proportion of forcing and warming that can be reliably attributed.



Figure 8: Projected N_2O concentrations, for SRES A2-marker scenario. Reference case (dotted curve); calculated with default emissions and response (solid curve); preferred case with emissions during historical period obtained by inversion of ice core data (dashed curve).

One complication that occurs for CFC-12 (CCl_2F_2) (and to a lesser extent other halocarbons) is the time delay between production (and application) and the emission through leakage during use and disposal.

For our preferred case, we use only CFC-11 and CFC-12, and specify concentrations according to the 'Copenhagen-like' scenario given by the IPCC (Prather et al., 1996: Table 2.5d). Restricting consideration to CFC-11 and CFC-12 should be a good approximation when considering attribution over the historical period, but attribution over the 21st century will need to consider a more comprehensive suite of halocarbons.

5 Validation of forcing and climate

5.1 Forcing

As noted by the SBSTA experts' meeting in Bonn (SBSTA, 2001), validating models against observed temperatures requires that all sources of radiative forcing be included – a rather complex task. Apart from the greenhouse gases (and gases that indirectly affect the atmospheric budgets of greenhouse gases), the main influences are:

- direct radiative effects of aerosols this depends on the type of aerosol, with a three-way classification into sulfate, carbon and organic commonly used;
- the indirect effect of aerosols through their role as cloud condensation nuclei;
- changes in stratospheric and tropospheric ozone;
- variations in solar irradiance.

These represent a mixture of anthropogenic and natural influences.

The two difficulties are:

- there is, in many cases, poor knowledge of the relation between the radiative forcing and the proximate cause that drives the forcing;
- there is generally poor knowledge of the historical pattern of either the forcing or its driver.

IPCC technical paper no. 2 (Harvey et al., 1997) gives a number of parameterisations of the forcing in terms of drivers. These would provide a starting point for extending the present calculations.

There have been a number of attempts to use empirical fitting to combine estimated histories of forcing. Two such studies are those of Stott et al. (2001) and Forest et al. (2002). Important aspects of the methodology in each case are (a) fitting spatial patterns of response, not just global mean time histories (b) presenting the results in terms of multi-variate distributions of estimates.

5.2 Climate response

As noted in the previous section, uncertainties in the historical forcing severely limit the scope for using calculated vs. observed warming as a basis for choosing between alternative climate model responses. Any such studies need to be undertaken (as in the references cited above) as joint estimation of climate parameters and forcing parameters.

Beyond such studies, all that we can do is compare a number of different cases, including the default, the response from Hasselmann et al. (1993), the time-dependent response from Watterson (2000) and various cases reported by den Elzen et al. (1999). Two that we note are:

Default This is the HadCM3 response, described by equations (2.5a). Our 'preferred' case also uses (2.5a) since the aim is to reproduce the results of the HadCM3 model.



Figure 9: Contributions to radiative forcing for default case.



Figure 10: Contributions to radiative forcing for 'preferred' case.



Figure 11: Projected warming, for SRES A2-marker scenario. Reference case (dotted curve); calculated with default emissions and response (solid curve); calculated with preferred model (dashed curve).

ECHAM-1 This is the single exponential response function given by Hasselmann et al. (1993). We have used this in our previous studies (Enting, 1998; Enting and Law, 2002).

Results for the default response are shown in Figure 11. It can be seen that, with the 'preferred' representations of greenhouse gases, the temperature curve gives an improved fit to the reference case.

Possible extensions to the analysis of the warming response are:

- Re-parameterise the time-dependent response given by Watterson (2000), so that nonlinearity is included explicitly rather than being incorporated through time as a proxy variable.
- Compared to other representations of climate response, equation (2.5a) combines a short time and a very long time, with no multi-decadal component. This is different from many other parameterisations. An explicit consequence is in the emission-time attribution calculations described in Section 6. Compared to the corresponding plots given by Enting and Law (2002), which were calculated with the ECHAM-1 response, the HadCM3 case has no un-realised warming on decadal time scales. This is likely to lead to significant differences in the relative attribution for groups of nations having distinct time histories of emissions, and so further studies of this issue are essential.



Figure 12: Projected sea-level rise: from reference (dotted curve); with default model (solid curve); with preferred model (dashed curve).

6 Emission-time attribution

The 'default' cases shown in Figures 3, 5 and 7 illustrate the importance of the emissions prior to 1890. While the fit can be improved by back-extrapolation of emissions, this is essentially empirical fitting and provides little basis for partitioning these emissions between national groups. Reliable information at the national level generally begins much later.

The SBSTA intercomparison specifies 3 start dates: 1890, 1950 and 1990. It also specifies calculations with emission end dates of 1990, 2000, 2050 and 2100, with attribution calculations performed for 2000, 2050, 2100 and 2500. We denote the three times involved as t_1 , the start time, t_2 the end time for emissions and t_a , the attribution time. In order to clarify the issues involved, we present (for the default case) the type of 'time-sliced' plot discussed by Enting and Law (2002). The various curves annotated with particular years give the amount of warming due to all emissions prior to that year. Following Enting and Law, we use the notation $W(t_a, t_2)$ to denote these curves, where the horizontal axis corresponds to t_a and the curves are for the specific values of t_2 as shown on the plot. To find the warming, at time t_a , due to emissions starting at t_1 and ending at t_2 , we simply take the difference $W(t_a, t_2) - W(t_a, t_1)$, i.e. the difference, on the plot, at time t_a , between the t_2 and t_1 curves.

Enting and Law (2002) noted that for non-linear relations (e.g. the relations between concentrations and forcing) there could be inconsistencies between attribution partitioned by time and attribution partitioned by nations. This would occur with either proportional attribution or with the differential form proposed by Enting (1998). Enting and Law proposed an alternative attri-



Figure 13: Attribution of warming for default case in terms of time of emission. Solid curve as in Figure 11. Dashed curves are for CO_2 , CH_4 and N_2O emissions ceasing at 1890, 1950, 1990 and 2050. Aerosol forcing continues as shown in Figure 2.

bution formalism that, at the cost of some computational effort, was explicitly consistent with emission-time-attribution. Consistency with the emission-time-attribution approach is needed once the attribution is calculated using alternative start dates.

The role of aerosols adds a significant complication to this issue. Figure 13 shows the partitioning of warming according to time of emissions of gases, with the aerosol forcing continuing. It indicates that until a time somewhat after 1990, cessation of gas emissions while continuing aerosol forcing could lead to near-zero warming within a few decades. In contrast, Figure 14 shows cases in which the emissions and the aerosol forcing cease simultaneously. The most immediate consequence is a rapid warming, reflecting the 8-year component of the HadCM2 response (see equation 2.5b).

Hansen et al. (2000) noted the importance of aerosol forcing, and the linkage between CO_2 emissions and aerosols. While this linkage is strong for global aggregates, there are significant differences between nations in the degree of linkage between aerosol emissions and CO_2 emissions.



Figure 14: Attribution of warming for default case in terms of time of emission. Solid curve as in Figure 11. Dashed curves are for CO_2 , CH_4 and N_2O emissions and aerosol forcing all ceasing at 1890, 1950, 1990 and 2050.

7 The SBSTA experts' recommendations

The SBSTA assessment is based largely on recommendations from the Bonn experts' meeting (SBSTA, 2001). While the present set of calculations merely represent an initial validation stage, it is important to review the recommendations to identify what subsequent steps are needed. The specific recommendations (using the paragraph numbering from SBSTA, 2001) for calculations were (in paraphrased form):

- 26 to use (and compare) a range of different indices for defining responsibility;
- 28 to present results in terms of a variety of socio-economic indicators;
- 32 uncertainties in attribution should be estimated;
- **34** the SRES scenarios should be used to examine sensitivity of attribution to scenario differences;
- 36 all greenhouse gases should be included for future periods;
- 37 calculations should assess the sensitivity to uncertainties in historic data, with various choices of start date;
- **38** a range of carbon cycle models should be used;

- 47 the sensitivity to differences in temperature responses should be examined;
- 49 the sensitivity of parameterisations of sea-level rise should be investigated.

The experts also noted a number of other matters that were important for the calculations. These included:

- **41,44** the need to include all contributions to radiative forcing to allow for validation against observed temperatures;
- 45 the need to include non-linearities in the relations between concentrations and forcing;

48 the need to include all contributions to sea-level rise.

It was also recommended [paragraph 30] that the code for participating models should be available.

Given these recommendations, we need to assess our modelling in terms of its ability to address them. Overall, the highly modularised form that we have adopted provides a good basis for performing the requisite sensitivity studies. It has been possible to define a 'preferred' case that is sufficiently close to the reference case from a full AOGCM, that alternative parameterisations (and the associated uncertainty ranges) can be evaluated in terms of whether the agreement with the reference remains satisfactory.

8 Sensitivity analysis

In studying the sensitivities in attribution calculations it is extremely important to distinguish two different aspects:

- the extent to which the results are sensitive to uncertainties in the science;
- the extent to which the results are sensitive to the choice of methodology.

A number of sensitivities related to each of the classes have been investigated by den Elzen et al. (1999) and more comprehensively by den Elzen and Schaeffer (2000, 2002).

They mainly considered the relative attribution to Annex-1 and others, and identified most important scientific uncertainty as:

• the size of the carbon flux from land-use change and its influence on the carbon budget.

They found that the greatest sensitivity to choice of methodology was (in order of decreasing importance):

- the inclusion of all CO₂ sources vs. only fossil CO₂;
- the inclusion of all gases vs. only CO₂;
- taking account of non-linearities in radiative forcing.

Characterising the uncertainty in parametric models raises the question of the appropriate parameter space to use. In many cases the parameter space required for specifying uncertainties is likely to be larger than the minimal parameter space required to specify the model. Indeed a 'full' characterisation of model uncertainty may well require a very large parameter space. However, for the SBSTA exercise, the requirements are, fortunately, much more specific and achievable. What is required is a knowledge of those aspects of model uncertainty that contribute to uncertainty in attribution. As noted in Section 3.2, we are really interested in a functional of the model and therefore only need consider sensitivities and uncertainties within a restricted space.

In exploring the ranges of uncertainty from models, the primary constraints come from observations. One needs to determine uncertainty ranges subject to acceptable (but not necessarily exact) agreement with observations in a complex, multi-parameter system. Our previous experience is that a combination of Bayesian calibration, combined with a Lagrange multiplier formalism for exploring extrema of functionals, provides an effective numerical technique for performing such studies (e.g. Enting and Pearman, 1987; Enting and Lassey, 1993).

9 Conclusions

The results presented in this report were calculated using a newly-developed modelling framework based on an object-oriented programming approach. Defining 'objects' with, so far as possible, standard interface characteristics, has enabled us to explore a range of different combinations of simple model components. The emphasis has been on addressing the specifications of the SBSTA intercomparison — demonstrating the ability of the simple (parametric) models to reproduce the behaviour of more complex (mechanistic) models. Conversely, in cases where the simple 'default' modelling approach appears incapable of reproducing the behaviour of the more complex models, relatively simple refinements have been identified, enabling our models to reproduce the specified behaviour.

The results of the comparisons can be summarised in a few general points:

i Matching future concentrations calculated by a more complicated model requires the use of a simple model with non-linear behaviour. Non-linearities in the uptake of CO_2 become increasingly important over the 21st century, as do, to a lesser extent, non-linearities in the response of CH_4 and N_2O .

- **ii** If the concentrations over the 21st century are in agreement with those obtained using a more complicated model then the warming matches that of the AOGCM (if using the default response from the AOGCM!!).
- iii The result that simple models can capture the behaviour of more complex models is hardly surprising. Such models have been used in this way for many years, and their use by the IPCC has been the subject of a special technical paper (Harvey et al., 1997). What remains to be demonstrated is the relation between the parameter space required to reproduce the behaviour and the parameter space required to characterise the uncertainties.
- iv The emission-time-attribution calculations reveal some of the factors discussed by Enting and Law (2002). Comparing the '1890', '1950' and '1990' curves shows that, over much of the 21st century, the attribution of warming to emissions from the 1890–1950 vs. 1950–1990 periods is, approximately, in the ratio 1:2. This confirms that information about pre-1950 sources is important, even if warming from such sources is not attributed. In analysing such periods concentration data from ice-cores (and in some cases from the CSIRO air archive) will be important. However, even with such data, the uncertainties regarding current atmospheric budgets of greenhouse gases are likely to be a major contribution to uncertainties in attribution.

Acknowledgements

The parameterisations of the HadCM3 model were supplied for the SBSTA intercomparison by Jason Lowe. The authors wish to thank Peter Rayner and John Garratt for valuable comments on the manuscript, and David Etheridge for advice on the ice-core data.

References

URLs are correct as of July 2002. Many of the UNFCCC documents are also available in languages other than English.

- van Aardenne, J.A., Dentener, F.J., Olivier, J.G.J., Klein Goldewijk, C.G.M. and Lelieveld, J. (2001) A $1^{\circ} \times 1^{\circ}$ resolution data set of historical anthropogenic trace gas emissions for the period 1890–1990. *Global Biogeochemical Cycles*, **15**, 909–928.
- Allen, L.H., Boote, K.J., Jones, J.W., Jones, P.H., Valle, R.R., Acock, B., Roger, H.H. and Dahlman, R.C. (1987) Response of vegetation to rising carbon dioxide: Photosynthesis, biomass and seed yield of soybean. *Global Biogeochemical Cycles*, **1**, 1–14.
- Anderssen, R.S. (1977) Some Numerical Aspects of Improperly Posed Problems: Why Regularization Works and When Not to Use It. Australian National University Computer Centre. Technical Report no. 52 (ANU: Canberra).
- Bacastow, R.B. (1981) Numerical evaluation of the evasion factor. pp95–101 of *Carbon Cycle Modelling; SCOPE 16.* Ed. B. Bolin (John Wiley and Sons: Chichester).
- Battle, M., Bender, M., Sowers, T., Tans, P.P., Butler, J.H., Ellis, J.T., Conway, T.J., Zhang, N., Lang, P. and Clarke, A.D. (1996) Atmospheric gas concentrations over the past century in air from firn at the South Pole. *Nature*, **383**, 231–235.
- Dai, A. and Fung, I.Y. (1993) Can climate variability contribute to the missing CO₂ sink? *Global Biogeochemical Cycles*, **7**, 599–609.
- den Elzen, M. and Schaeffer, M. (2000) Assessment of Major Uncertainties in Calculating Regional Contributions of Climate Change. RIVM Report 728001012. (Rijksinstituut voor Volksgezondheid en Milieu: Bilthoven).

- den Elzen, M. and Schaeffer, M. (2002) Responsibility for past and future global warming: Uncertainties in attributing anthropogenic climate change. *Climatic Change*, **54**, 29–73.
- den Elzen, M. Berk, M., Schaeffer, M., Olivier, J., Hendriks, C. and Metz, B. (1999) The Brazilian Proposal and Other Options for International Burden Sharing: An Evaluation of Methodological and Policy Aspects Using the FAIR Model. RIVM Report 728001011. (Rijksinstituut voor Volksgezondheid en Milieu: Bilthoven). http://unfccc.int/sessions/workshop/010528/rap728001011.pdf
 - http://www.rivm.nl:80/bibliotheek/rapporten/728001011.html
- Enting, I.G. (1990) Ambiguities in the calibration of carbon cycle models. *Inverse Problems*, **6**, L39–46.
- Enting, I.G. (1991) *Calculating Future Atmospheric CO*₂ *Concentrations*. CSIRO Division of Atmospheric Research Technical Paper no. 22. (CSIRO: Australia). Also electronic edition (pdf),

at http://www.dar.csiro.au/publications/Enting_2000d.pdf

- Enting, I.G. (1998) Attribution of Greenhouse Gas Emissions, Concentrations and Radiative Forcing. CSIRO Atmospheric Research Technical Paper no. 38. (CSIRO: Australia). Also electronic edition (636 kB pdf),
- at http://www.dar.csiro.au/publications/Enting_2000c.pdf
- Enting, I.G. (2002) Inverse Problems in Atmospheric Constituent Transport. (CUP: Cambridge).

http://www.rivm.nl:80/bibliotheek/rapporten/728001012.pdf

- Enting, I.G. and Lassey, K.R. (1993) *Projections of Future CO*₂. CSIRO Division of Atmospheric Research Technical Paper no. 27. (CSIRO: Australia). Also electronic edition (pdf), at *http://www.dar.csiro.au/publications/Enting_2000e.pdf*
- Enting, I.G. and Law, R.M. (2002) *Characterising Historical Responsibility for the Greenhouse Effect.* CSIRO Atmospheric Research Technical Paper no. 41. (CSIRO: Australia). Also electronic edition (pdf),

at http://www.dar.csiro.au/publications/Enting_2002a.pdf

- Enting, I.G. and Pearman, G.I. (1987) Description of a one-dimensional carbon cycle model calibrated using techniques of constrained inversion. *Tellus*, **39B**, 459–476.
- Enting, I.G. and Trudinger, C.M. (2001) Quantifying uncertainties in projections of future CO₂. pp 801–804 of *Extended Abstracts: 6th International Carbon Dioxide Conference*. (vol II). (Organizing committee: Tohoku University, Sendai, Japan)
- Enting, I.G., Wigley, T.M.L. and Heimann, M. (1994) Future Emissions and Concentrations of Carbon Dioxide: Key Ocean/Atmosphere/Land Analyses. CSIRO Division of Atmospheric Research Technical Paper no. 31. (CSIRO: Australia).
 Also electronic edition (HTML index to PDF files for sections),

at http://www.dar.csiro.au/publications/Enting_2001a0.htm

- Etheridge, D.M., Steele, L.P., Langenfelds, R.L., Francey, R.J., Barnola, J.-M., and Morgan, V.I. (1996) Natural and anthropogenic changes in atmospheric CO₂ over the last 1000 years from air in Antarctic ice and firn. *J. Geophys. Res.*, **101D**, 4115–4128.
- Etheridge, D.M., Steele, L.P., Francey, R.J. and Langenfelds, R.L. (1998) Atmospheric methane between 1000 A.D. and present: Evidence of anthropogenic emissions and climatic variability. J. Geophys. Res., 103D, 15979–15993.
- Forest, C.E., Stone, P.H., Sokolov, A.P., Allen, M.R. and Webster, M.D. (2002) Quantifying uncertainties in climate system properties with the use of recent climate observations. *Science*, 295, 113–117.
- Hansen, J., Sato, M., Ruedy, R., Lacis, A. and Oinas, V. (2000) Global warming in the twentyfirst century: an alternative scenario. *Proc. Nat. Acad. Sci.*, **97**, 9875–9880.
- Harvey, L.D.D. (2000) Global Warming: The Hard Science. (Prentice Hall: Harlow, U.K.).
- Harvey, D., Gregory, J., Hoffert, M., Jain, A., Lal, M., Leemans, R., Raper, S., Wigley, T. and de Wolde, J. (1997) An Introduction to Simple Climate Models used in the IPCC Second Assessment Report. IPCC Technical Paper II. (Intergovernmental Panel on Climate Change). http://www.ipcc.ch/pub/IPCCTP.II(E).pdf
- Hasselmann, K., Sausen, R., Maier-Reimer, E. and Voss, R. (1993) On the cold start problem in transient simulations with coupled atmosphere-ocean models. *Climate Dynamics*, **9**, 53–61.
- Houghton, R.A. (1999) The annual net flux of carbon to the atmosphere from changes in land use 1850–1990. *Tellus*, **51B**, 298–313.
- Houghton, R.A. and Hackler, J.L. (2000) Carbon flux to the atmosphere from land-use change. In *Trends-Online*.

http://cdiac.esd.ornl.gov/trends/landuse/houghton/houghton.html

- IPCC (2001) *Climate Change 2000: The Scientific Basis.* Ed. J.T. Houghton, Y. Ding, D.J. Griggs, M. Noguer, P.J. van der Linden, X. Dai, K. Maskell and C.A. Johnson. (Published for the Intergovernmental Panel on Climate Change by CUP: Cambridge, U.K.). http://www.ipcc.ch/pub/tar/wg1/index.htm
- Joos, F., Bruno, M., Fink, R., Siegenthaler, U., Stocker, T., Le Quéré, C. and Sarmiento, J.L. (1996) An efficient and accurate representation of complex oceanic and biospheric models

of anthropogenic carbon uptake. Tellus, 48B, 397-417.

- Kaminski, T., Rayner, P.J., Heimann, M. and Enting, I.G. (2001) On aggregation errors in atmospheric transport inversions. J. Geophys. Res., **106D**, 4703–4715.
- Machida, T., Nakazawa, T., Fujii, Y., Aoki, S. and Watanabe, O. (1995) Increase in the atmospheric nitrous oxide concentration during the last 250 years. *Geophys. Res. Lett.*, **22**, 2921–2924.
- Maier-Reimer, E. and Hasselmann, K. (1987) Transport and storage of CO₂ in the ocean an inorganic ocean-circulation carbon cycle model. *Climate Dynamics*, **2**, 63–90.
- Marland, G., Boden, T.A. and Andres, R.J. (2001) Global, regional and national fossil fuel CO₂ emissions. In *Trends: A Compendium of Data on Global Change*. (CDIAC, ORNL: Oak Ridge, Tennessee).

http://cdiac.esd.ornl.gov/trends/emis/em_cont.htm

- Meira Filho, L.G. and Miguez, J.G.D. (1998) Note on the Time Dependent Relationship Between Emissions of Greenhouse Gases and Climate Change. (Ministry of Science and Technology: Brasilia, Brazil). Electronic version (2000), English Language: http://www.mct.gov.br/clima/ingles/negoc/pdf/proposta.pdf
- Meyer, R., Joos, F., Esser, G., Heimann, M., Hooss, G., Kohlmaier, G., Sauf, W., Voss, R. and Wittenberg, U. (1999) The substitution of high-resolution terrestrial biosphere models and carbon sequestration in response to changing CO₂ and climate. *Global Biogeochemical Cycles*, **13**, 785–802.
- Nakićenović, Alcamo, J., Davis, G., de Vries, B., Fenham, J., Gaffin, S., Gregory, K., Grübler.A., Jung, T.Y., Kram, T., La Rovere, E.L., Michaelis, L., Mori, S., Morita, T., Pepper, W., Pitcher, H., Price, L., Riahi, K., Roehrl, A., Rogner, H.-H., Sankovski, A., Schlesinger, M., Shukla, P., Smith, S., Swart, R., van Rooijen, S., Victor, N. and Zhou, D. (2000) *Emission Scenarios*. (Published for the Intergovernmental Panel on Climate Change by CUP: Cambridge, U.K.).
- Prather, M.J. (1994) Lifetimes and eigenstates in atmospheric chemistry. *Geophys. Res. Lett.*, **21**, 801–804.
- Prather, M., Derwent, R., Ehhalt, D., Fraser, P., Sanhueza, E. and Zhou, X. (1996) Other trace gases and atmospheric chemistry. Section 2.2 of *Climate Change 1995: The Science of Climate Change.* Ed. J.T. Houghton, L.G. Meira Filho, B.A. Callander, N. Harris, A. Kattenberg and K. Maskell. (Published for the Intergovernmental Panel on Climate Change by CUP: Cambridge, U.K.).
- Prather, M., Ehhalt, D., Dentener, F., Derwent, R., Dlugokencky, E., Holland, E., Isaksen, I., Katima, J., Kirchhoff, V., Matson, P., Midgley, P., and Wang, M. (2001) Atmospheric chemistry and greenhouse gases. Chapter 4 of *Climate Change 2000: The Scientific Basis*. Ed. J.T. Houghton, Y. Ding, D.J. Griggs, M. Noguer, P.J. van der Linden, X. Dai, K. Maskell and C.A. Johnson. (Published for the Intergovernmental Panel on Climate Change by CUP: Cambridge, U.K.).
- Ramaswamy, V., Boucher, O., Haigh, J., Hauglustaine, D., Haywood, J., Myhre, G., Nakajima, T., Shi, G.Y. and Solomon, S. (2001) Radiative Forcing of Climate Change. pp349–416 of *Climate Change 2000: The Scientific Basis.* Ed. J.T. Houghton, Y. Ding, D.J. Griggs, M. Noguer, P.J. van der Linden, X. Dai, K. Maskell and C.A. Johnson. (Published for the Intergovernmental Panel on Climate Change by CUP: Cambridge, U.K.).
- SBSTA (2001) Scientific and Methodological Aspects of the Proposal by Brazil: Progress Report on the Review of the Scientific and Methodological Aspects of the Proposal by Brazil.

FCCC/SBSTA/2001/INF.2. (UNFCC: Bonn).

http://www.unfccc.int/resource/docs/2001/sbsta/inf02.pdf

- Shine, K.P., Derwent, R.G., Wuebbles, D.J. and Morcrette, J.-J. (1990) Radiative Forcing of Climate. In *Climate Change; The IPCC Scientific Assessment*. Ed. J.T Houghton, G.J. Jenkins and J.J. Ephraums. (Published for the Intergovernmental Panel on Climate Change by CUP: Cambridge, U.K.).
- Stott, P.A., Tett, S.F.B., Jones, G.S., Allen, M.R., Ingram, W.J. and Mitchell, J.F.B. (2001) Attribution of twentieth century temperature change to natural and anthropogenic causes. *Climate Dynamics*, **17**, 1–21.
- Thompson, M.V. and Randerson, J.T. (1999) Impulse response functions of terrestrial carbon cycle models: method and applications. *Global Change Biology*, **5**, 371–394.
- Trudinger, C.M. (2000) The Carbon Cycle Over the Last 1000 Years Inferred from Inversion of Ice-Core Data. Ph.D. Thesis, Monash University. Also electronic edition (HTML index to PDF files), at http://www.dar.csiro.au/publications/Trudinger_2001a0.htm
- Trudinger, C.M., Enting, I.G. and Rayner, P.J. (2002a) Kalman filter analysis of ice core data:

1. Method development and testing the statistics. J. Geophys. Res. (in press).

- Trudinger, C.M., Enting, I.G., Rayner, P.J. and Francey, R.J. (2002b) Kalman filter analysis of ice core data: 2. Double deconvolution of CO_2 and $\delta^{13}C$ measurements. *J. Geophys. Res.* (in press).
- Watterson, I.G. (2000) Interpretation of simulated global warming using a simple model. J. *Climate*, **13**, 202–215.
- Wigley, T.M.L. (1991) A simple inverse carbon cycle model. *Global Biogeochemical Cycles*, **5**, 373–382.
- Wigley, T.M.L. and Raper, S.C.B. (1992) Implications for climate and sea level rise of the revised IPCC emission scenarios. *Nature*, **357**, 293–300.

Notation

- a_{η} Radiative effect per unit of gas η in W m⁻² per unit mass. For GWPs, W m⁻² kg⁻¹ are used (c.f. Enting and Law, 2002).
- A_{η} Scale factor converting mass units to concentration units for gas η .
- C_0 Equilibrium CO₂ concentration in pre-industrial period.
- C_c Minimum CO₂ for photosynthesis.
- E(t) Emission rate, for CO₂ unless otherwise indicated.
- F(t) Radiative forcing.
- G_{∞} Maximum enhancement factor for NPP.
- $F_{\eta}(t)$ Radiative forcing from constituent η .
- $M_{\rm a}$ Atmospheric carbon content.

- $M_{\text{a:eqm}}$ Atmospheric carbon content at pre-industrial equilibrium.
- $M_{\rm m}$ Ocean mixed-layer carbon content.
- $M_{\rm m:eqm}$ Ocean mixed-layer carbon content at pre-industrial equilibrium.
- $M'_{\rm m}$ Increase in ocean mixed-layer carbon content above pre-industrial equilibrium.
- R(t) Response function. For CO₂ unless otherwise indicated.
- $R_{\eta}(t)$ Response function for gas η .
- S(t) Sea-level rise.
- t Time (in years).
- t_0 Initial time, before which net anthropogenic emissions were taken as zero.
- t_1 Initial time, when considering anthropogenic emissions over a fixed interval $[t_1, t_2]$.
- t_2 Final time, when considering anthropogenic emissions over a fixed interval $[t_1, t_2]$.
- t_a Time to which attribution calculations apply.
- T Temperature increase.
- W(t) Response function for temperature increase driven by radiative forcing.
- W(t, t') Warming at time t due to emissions prior to time t'.
- V(t) Response function for sea-level rise driven by temperature changes.
- $\Phi_{\rm AS}$ Air-sea carbon flux.
- $\Phi_{\rm NPP}$ Net primary production.
- $\Phi'_{\rm NPP}$ Increase in net primary production relative to equilibrium.
- η (as subscript) generic subscript for greenhouse gases.
- κ Air-sea gas exchange rate, as inverse of atmospheric turnover time.
- τ_{η} Lifetime for gas η .
- ξ Buffer factor.

Results submitted for SBSTA Assessment (phase 1)

Default case

Date	CO_2	CH_4	N_2O	Forcing	Temperature	Sea-level
	ppm	ppb	ppb	${ m W}~{ m m}^{-2}$	K	m
1750.0	280.00	700.0	275.00	0.0000	0.0000	0.0000
1760.0	280.07	708.3	275.03	0.0082	0.0017	0.0000
1770.0	280.25	724.6	275.11	0.0250	0.0084	0.0001
1780.0	280.53	743.4	275.24	0.0454	0.0191	0.0004
1790.0	280.88	762.9	275.42	0.0680	0.0323	0.0007
1800.0	281.30	782.7	275.64	0.0920	0.0472	0.0013
1810.0	281.81	802.5	275.90	0.1175	0.0633	0.0019
1820.0	282.38	822.3	276.20	0.1442	0.0806	0.0028
1830.0	283.02	842.2	276.53	0.1722	0.0989	0.0038
1840.0	283.75	862.0	276.89	0.2017	0.1184	0.0051
1850.0	284.57	881.9	277.27	0.2329	0.1391	0.0065
1860.0	285.50	901.8	277.68	0.2338	0.1591	0.0081
1870.0	286.53	921.6	278.12	0.2347	0.1608	0.0096
1880.0	287.85	941.5	278.58	0.2417	0.1646	0.0111
1890.0	289.59	961.3	279.06	0.2553	0.1720	0.0127
1900.0	291.58	989.1	279.55	0.2795	0.1842	0.0144
1910.0	294.58	1023.4	280.09	0.3012	0.1992	0.0162
1920.0	297.61	1061.5	280.72	0.3241	0.2164	0.0182
1930.0	300.83	1099.9	281.42	0.3510	0.2340	0.0203
1940.0	303.97	1137.2	282.16	0.3738	0.2519	0.0226
1950.0	307.95	1182.4	283.17	0.4169	0.2750	0.0251
1960.0	315.11	1269.3	284.61	0.4044	0.2862	0.0276
1970.0	325.66	1373.8	286.60	0.4554	0.3044	0.0303
1980.0	339.52	1473.9	289.54	0.6185	0.3579	0.0335
1990.0	355.44	1569.3	293.47	0.8686	0.4803	0.0383
2000.0	369.41	1620.5	305.26	1.0805	0.6235	0.0445
2010.0	386.40	1704.8	317.48	1.3321	0.7757	0.0522
2020.0	408.57	1836.2	331.24	1.6679	0.9632	0.0618
2030.0	436.12	1997.5	346.42	2.0696	1.1978	0.0738
2040.0	466.12	2167.2	362.02	2.4789	1.4640	0.0885
2050.0	497.21	2334.4	377.64	2.8726	1.7367	0.1058
2060.0	530.35	2500.9	393.55	3.4191	2.0514	0.1261
2070.0	566.00	2668.4	410.04	3.9670	2.4102	0.1498
2080.0	605.06	2839.5	427.05	4.5237	2.7877	0.1771
2090.0	649.09	3013.4	444.44	5.0981	3.1808	0.2080
2100.0	698.28	3189.3	462.08	5.6861	3.5892	0.2426

Preferred case

Date	CO_2	CH_4	N_2O	Forcing	Temperature	Sea-level
	ppm	ppb	ppb	${ m W}~{ m m}^{-2}$	K	m
1750.0	282.50	700.0	275.00	0.0000	0.0000	0.0000
1760.0	282.57	708.3	275.02	0.0081	0.0017	0.0000
1770.0	282.74	724.6	275.01	0.0243	0.0082	0.0001
1780.0	282.99	743.4	274.95	0.0436	0.0185	0.0003
1790.0	283.30	762.9	274.85	0.0644	0.0309	0.0007
1800.0	283.68	782.7	274.72	0.0863	0.0446	0.0012
1810.0	284.12	802.5	274.58	0.1091	0.0592	0.0018
1820.0	284.61	822.3	274.47	0.1329	0.0747	0.0026
1830.0	285.16	842.2	274.44	0.1578	0.0911	0.0036
1840.0	285.78	862.0	274.50	0.1842	0.1085	0.0047
1850.0	286.49	881.9	274.68	0.2124	0.1271	0.0060
1860.0	287.28	901.8	274.99	0.2104	0.1451	0.0074
1870.0	288.16	921.6	275.42	0.2083	0.1447	0.0088
1880.0	289.31	941.5	275.99	0.2124	0.1464	0.0101
1890.0	290.85	961.3	276.73	0.2230	0.1517	0.0115
1900.0	292.60	989.1	277.69	0.2444	0.1617	0.0130
1910.0	295.30	1024.6	278.84	0.2632	0.1746	0.0146
1920.0	297.99	1065.5	280.69	0.2859	0.1903	0.0163
1930.0	300.82	1107.7	282.27	0.3112	0.2070	0.0182
1940.0	303.57	1149.5	282.87	0.3292	0.2228	0.0202
1950.0	307.11	1200.1	284.31	0.3696	0.2431	0.0224
1960.0	313.77	1294.1	288.38	0.3615	0.2538	0.0247
1970.0	323.76	1408.7	293.16	0.4718	0.2883	0.0272
1980.0	337.06	1520.9	300.72	0.7323	0.3829	0.0310
1990.0	352.52	1630.4	308.05	1.0753	0.5607	0.0368
2000.0	366.19	1695.7	318.68	1.3150	0.7462	0.0445
2010.0	383.19	1794.7	329.83	1.5450	0.9118	0.0537
2020.0	406.11	1945.3	342.60	1.8709	1.0981	0.0646
2030.0	435.71	2131.7	356.87	2.2833	1.3347	0.0780
2040.0	469.41	2331.8	371.64	2.7233	1.6130	0.0942
2050.0	505.95	2533.3	386.48	3.1647	1.9091	0.1132
2060.0	546.73	2737.4	401.69	3.7752	2.2581	0.1356
2070.0	592.80	2945.7	417.53	4.4052	2.6627	0.1619
2080.0	645.76	3161.0	433.94	5.0622	3.0979	0.1924
2090.0	708.25	3382.7	450.78	5.7569	3.5615	0.2272
2100.0	781.73	3610.1	467.92	6.4855	4.0541	0.2665

These results were submitted on June 6 2002, to replace earlier results that used an incorrect implementation of the HadCM2 responses. These revisions also remove minor discrepancies between the original calculations and the specifications given in this report.

CSIRO Atmospheric Research Technical Papers

This series has been issued as *Division of Atmospheric Research Technical Paper* (nos. 1–19); *CSIRO Division of Atmospheric Research Technical Paper* (nos. 20–37) and *CSIRO Atmospheric Research Technical Paper* from no. 38.

Regular electronic publication commenced with no. 45. Earlier technical papers are progressively being made available in electronic form. A current list of technical papers is maintained on the CAR website, currently at *http://www.dar.csiro.au/information/TP.htm.* Papers may be issued out of sequence.

- No. 1 Galbally, I.E.; Roy, C.R.; O'Brien, R.S.; Ridley, B.A.; Hastie, D.R.; Evans, W.J.F.;
 McElroy, C.T.; Kerr, J.B.; Hyson, P.; Knight, W.; Laby, J.E. *Measurements of trace composition of the Austral stratosphere: chemical and meteorological data*. 1983. 31
 p.
- No. 2 Enting, I.G. *Error analysis for parameter estimates from constrained inversion*. 1983. 18 p.
- No. 3 Enting, I.G.; Pearman, G.I. *Refinements to a one-dimensional carbon cycle model*. 1983. 35 p.
- No. 4 Francey, R.J.; Barbetti, M.; Bird, T.; Beardsmore, D.; Coupland, W.; Dolezal, J.E.;
 Farquhar, G.D.; Flynn, R.G.; Fraser, P.J.; Gifford, R.M.; Goodman, H.S.; Kunda, B.;
 McPhail, S.; Nanson, G.; Pearman, G.I.; Richards, N.G.; Sharkey, T.D.; Temple, R.B.;
 Weir, B. *Isotopes in tree rings.* 1984. 86 p.
- No. 5 Enting, I.G. Techniques for determining surface sources from surface observations of atmospheric constituents. 1984. 30 p.
- No. 6 Beardsmore, D.J.; Pearman, G.I.; O'Brien, R.C. *The CSIRO (Australia) Atmospheric Carbon Dioxide Monitoring Program: surface data*. 1984. 115 p.
- No. 7 Scott, John C. *High speed magnetic tape interface for a microcomputer.* 1984. 17 p.
- No. 8 Galbally, I.E.; Roy, C.R.; Elsworth, C.M.; Rabich, H.A.H. *The measurement of nitrogen oxide* (*NO*, *NO*₂) *exchange over plant/soil surfaces*. 1985. 23 p.
- No. 9 Enting, I.G. A strategy for calibrating atmospheric transport models. 1985. 25 p.
- No. 10 O'Brien, D.M. TOVPIX: software for extraction and calibration of TOVS data from the high resolution picture transmission from TIROS-N satellites. 1985. 41 p.
- No. 11 Enting, I.G.; Mansbridge, J.V. Description of a two-dimensional atmospheric transport model. 1986. 22 p.
- No. 12 Everett, J.R.; O'Brien, D.M.; Davis, T.J. A report on experiments to measure average fibre diameters by optical fourier analysis. 1986. 22 p.
- No. 13 Enting, I.G. A signal processing approach to analysing background atmospheric constituent data. 1986. 21 p.
- No. 14 Enting, I.G.; Mansbridge, J.V. *Preliminary studies with a two-dimensional model using transport fields derived from a GCM*. 1987. 47 p.
- No. 15 O'Brien, D.M.; Mitchell, R.M. Technical assessment of the joint CSIRO/Bureau of Meteorology proposal for a geostationary imager/sounder over the Australian region. 1987. 53 p.
- No. 16 Galbally, I.E.; Manins, P.C.; Ripari, L.; Bateup, R. A numerical model of the late

(ascending) stage of a nuclear fireball. 1987. 89 p.

- No. 17 Durre, A.M.; Beer, T. *Wind information prediction study: Annaburroo meteorological data analysis.* 1989. 30 p. + diskette.
- No. 18 Mansbridge, J.V.; Enting, I.G. Sensitivity studies in a two-dimensional atmospheric transport model. 1989. 33 p.
- No. 19 O'Brien, D.M.; Mitchell, R.M. Zones of feasibility for retrieval of surface pressure from observations of absorption in the A band of oxygen. 1989. 12 p.
- No. 20 Evans, J.L. Envisaged impacts of enhanced greenhouse warming on tropical cyclones in the Australian region. 1990. 31 p. [Out of print]
- No. 21 Whetton, P.H.; Pittock, A.B. Australian region intercomparison of the results of some general circulation models used in enhanced greenhouse experiments. 1991. 73 p. [Out of print]
- No. 22 Enting, I.G. *Calculating future atmospheric CO*₂ *concentrations*. 1991. 32 p. Also electronic edition (718 kB pdf), at http://www.dar.csiro.au/publications/Enting_2000d.pdf
- No. 23 Kowalczyk, E.A.; Garratt, J.R.; Krummel, P.B. A soil-canopy scheme for use in a numerical model of the atmosphere 1D stand-alone model. 1992. 56 p.
- No. 24 Physick, W.L.; Noonan, J.A.; McGregor, J.L.; Hurley, P.J.; Abbs, D.J.; Manins, P.C. *LADM: A Lagrangian Atmospheric Dispersion Model.* 1994. 137 p.
- No. 25 Enting, I.G. *Constraining the atmospheric carbon budget: a preliminary assessment.* 1992. 28 p. Also electronic edition (571 kB), at http://www.dar.csiro.au/publications/Enting_2000b.pdf
- No. 26 McGregor, J.L.; Gordon, H.B.; Watterson, I.G.; Dix, M.R.; Rotstayn, L.D. *The CSIRO* 9-level atmospheric general circulation model. 1993. 89 p.
- No. 27 Enting, I.G.; Lassey, K.R. *Projections of future CO*₂. with appendix by R.A. Houghton. 1993. 42 p. Also electronic edition (860 kB pdf), at http://www.dar.csiro.au/publications/Enting_2000e.pdf
- No. 28 [Not published]
- No. 29 Enting, I.G.; Trudinger, C.M.; Francey, R.J.; Granek, H. Synthesis inversion of atmospheric CO₂ using the GISS tracer transport model. 1993. 44 p.
- No. 30 O'Brien, D.M. Radiation fluxes and cloud amounts predicted by the CSIRO nine level GCM and observed by ERBE and ISCCP. 1993. 37 p.
- No. 31 Enting, I.G.; Wigley, T.M.L.; Heimann, M. Future emissions and concentrations of carbon dioxide: key ocean/atmosphere/land analyses. 1993. 120 p. Also electronic edition (HTML index to pdf files), at http://www.dar.csiro.au/publications/Enting_2001a0.htm
- No. 32 Kowalczyk, E.A.; Garratt, J.R.; Krummel, P.B. *Implementation of a soil-canopy scheme*
- into the CSIRO GCM regional aspects of the model response. 1994. 59 p.
- No. 33 Prata, A.J. Validation data for land surface temperature determination from satellites. 1994. 36 p.
- No. 34 Dilley, A.C.; Elsum, C.C. *Improved AVHRR data navigation using automated land feature recognition to correct a satellite orbital model.* 1994. 22 p.
- No. 35 Hill, R.H.; Long, A.B. The CSIRO dual-frequency microwave radiometer. 1995. 16 p.
- No. 36 Rayner, P.J.; Law, R.M. A comparison of modelled responses to prescribed CO₂ sources. 1995. 84 p.
- No. 37 Hennessy, K.J. CSIRO Climate change output. 1998. 23 p.

- No. 38 Enting, I.G. Attribution of greenhouse gas emissions, concentrations and radiative forcing. 1998. 27 p. Also electronic edition (557 kB pdf), at http://www.dar.csiro.au/publications/Enting_2000c.pdf
- No. 39 O'Brien, D.M.; Tregoning, P. Geographical distributions of occultations of GPS satellites viewed from a low earth orbiting satellite. (1998) 28p.
- No. 40 Enting, I.G. Characterising the temporal variability of the global carbon cycle. 1999.
 53 p. Also electronic edition (636 kB pdf), at http://www.dar.csiro.au/publications/Enting_2000a.pdf
- No. 41 Enting, I.G.; Law, R.M. Characterising historical responsibility for the greenhouse effect. 2002.
 Electronic edition only (pdf), at http://www.dar.csiro.au/publications/enting_2002.pdf
- No. 42 Mitchell, R.M. Calibration status of the NOAA AVHRR solar reflectance channels: CalWatch revision 1. 1999. 20 p.
- No. 43 Hurley, P.J. *The Air Pollution Model (TAPM) Version 1: technical description and examples.* 1999. 41 p. Also electronic edition (276 kB pdf), at http://www.dar.csiro.au/publications/Hurley_1999a.pdf
- No. 44 Frederiksen, J.S.; Dix, M.R.; Davies, A.G. A new eddy diffusion parameterisation for the CSIRO GCM. 2000. 31 p.
- No. 45 Young, S.A. Vegetation Lidar Studies. Electronic edition in preparation.
- No. 46 Prata, A. J. Global Distribution of Maximum Land Surface Temperature Inferred from Satellites: Implications for the Operation of the Advanced Along Track Scanning Radiometer. 2000. 30 p. Electronic edition only (8307 kB pdf), at http://www.dar.csiro.au/publications/Prata_2000a.pdf
- No. 47 Prata, A. J. *Precipitable water retrieval from multi-filter rotating shadowband radiometer measurements.* 2000. 14 p. Electronic edition only (1554 kB pdf), at http://www.dar.csiro.au/publications/Prata_2000b.pdf
- No. 48 Prata, A.J.; Grant, I.F. *Determination of mass loadings and plume heights of volcanic ash clouds from satellite data.* 2001. 39p. Electronic edition only (pdf), at http://www.dar.csiro.au/publications/Prata_2001a.pdf
- No. 49 O'Brien, D.M. Numerical calculation of the transfer of polarized radiation by a scattering and absorbing atmosphere. 2001. Electronic edition only (pdf), at http://www.dar.csiro.au/publications/O'Brien_2001a.pdf
- No. 50 Law, R.M.; Vohralik, P.F. *Methane sources from mass-balance inversions: Sensitivity to transport.* 2001. 27p. Electronic edition only (pdf), at http://www.dar.csiro.au/publications/Law_2001a.pdf
- No. 51 Meyer, C.P.; Galbally, I.E.; Wang, Y.-P.; Weeks, I.A.; Jamie, I.; Griffith, D.W.T. *Two* automatic chamber techniques for measuring soil-atmosphere exchanges of trace gases and results of their use in the OASIS field experiment. 2001. 33p. Electronic edition only (pdf),

at http://www.dar.csiro.au/publications/Meyer_2001a.pdf

- No. 52 Mitchell, R.M. *In-flight characteristics of the space count of NOAA AVHRR channels 1 and 2.* 2001. 24p. Electronic edition only (pdf), at http://www.dar.csiro.au/publications/MitchellR_2001a.pdf
- No. 53 Young, S.A. An investigation into the performance of algorithms used to retrieve cloud parameters from LITE lidar data, and implications for their use with PICASSO-CENA

lidar data. 2001. Electronic edition only (pdf), at *http://www.dar.csiro.au/publications/Young_2001a.pdf*

- No. 54 [in preparation]
- No. 55 Hurley, P. *The Air Pollution Model (TAPM) Version 2. Part 1. Technical description.* 2002. Electronic edition only (pdf), at http://www.dar.csiro.au/publications/tapm_v2_technical_paper_part1.pdf
- No. 56 Enting, I.G.; Trudinger, C.M. *Modelling earth system change. I. Validating parameterisations for attribution calculations.* 2002. 36p. Electronic edition only (pdf), at http://www.dar.csiro.au/publications/enting_2002b.pdf
- No. 57 Hurley, P. *The Air Pollution Model (TAPM) Version 2. Part 2. Summary of some verification studies.* 2002. Electronic edition only (pdf), at http://www.dar.csiro.au/publications/hurley_2002b.pdf

Address and contact details: CSIRO Atmospheric Research Private Bag No.1 Aspendale Victoria 3195 Australia Ph: (+61 3) 9239 4400; fax: (+61 3) 9239 4444 e-mail: ar-enquiries@csiro.au

Last updated 18 July 2002.